Author's response to reviews

Title: Prevalence of methicillin-resistant Staphylococcus aureus colonization in HIV-infected patients in Barcelona, Spain: A cross-sectional study

Authors:

Arkaitz Imaz (aimaz@bellvitgehospital.cat)
Mariana Camoez (mcamoez@bellvitgehospital.cat)
Silvana Di Yacovo (mdiycovo@bellvitgehospital.cat)
Oriol Gasch (ogasch@bellvitgehospital.cat)
M Angeles Dominguez (adominguez@bellvitgehospital.cat)
Antonia Vila (avila@bellvitgehospital.cat)
Margarita Maso-Serra (mmaso@bellvitgehospital.cat)
Miquel Pujol (mpujol@bellvitgehospital.cat)
Daniel Podzamczer (dpodzamczer@bellvitgehospital.cat)

Version: 4 Date: 14 May 2015

Author's response to reviews: see over
Barcelona, 14 May 2015

Philippa Harris,
Executive Editor, BioMed Central

**Manuscript ID:** 7534701411590112 ("Prevalence of Community-Associated methillin-resistant Staphylococcus aureus colonization in HIV-infected patients in Barcelona, Spain")

Thank you for your interest in our manuscript and the opportunity to publish our research in BMC Infectious Diseases. We are now submitting a revised version incorporating the changes that the editors have requested.

We hope that in its present form you will find our manuscript suitable for publication and look forward to your reply.

Yours sincerely,

Arkaitz Imaz, MD, PhD
HIV Unit, Infectious Diseases Department
Hospital Universitari de Bellvitge
Barcelona, SPAIN
aimaz@bellvitgehospital.cat;

1. **Please could you rename the title "Introduction" to "Background".**
   Yes. It has been done.
2. The Authors contributions section needs to be a separate section before the Acknowledgements and after the Conflict of Interest statement.

Ok. We have changed it.